Author's response to reviews

Title: The cost effectiveness of an early transition from hospital to nursing home for stroke patients: design of a comparative study

Authors:

Ron WH Heijnen (R.heijnen@umc.nl)
Silvia MAA Evers (S.Evers@BEOZ.unimaas.nl)
Trudy DEM van der Weijden (Trudy.vanderWeijden@hag.unimaas.nl)
Martien Limburg (martienlimburg@gmail.com)
Jos MGA Schols (Jos.Schols@hag.unimaas.nl)

Version: 7 Date: 26 May 2010

Author's response to reviews:

Maastricht, 2010-05-26

Dear Editor,

there has been one final revision made in the text.

In the discussion on pag 17, the word "may" has been replaced by "will"
in the sentence

The results of this study may provide information about the cost effectiveness of
the intervention and its effects on clinical outcomes and quality of life.

Kind regards